Author's response to reviews

Title: Improvement of primary care for patients with chronic heart failure: a pilot study.

Authors:

Jan van Lieshout (j.vanlieshout@iq.umcn.nl)
Michel Wensing (m.wensing@iq.umcn.nl)
Richard Grol (r.grol@iq.umcn.nl)

Version: 3 Date: 4 December 2009

Author's response to reviews: see over
Dear Editor,

Herewith we submit our revised manuscript entitled ‘Improvement of primary care for patients with chronic heart failure: a pilot study’.

Considering the remark about language corrections needed we had the manuscript checked critically and we could make several adjustments throughout the text. Furthermore, we present the name of the ethical committee. We hope these changes improve our article and that you will consider it ready for publication.

Yours sincerely, on behalf of all the authors,

Jan van Lieshout
General practitioner researcher

Scientific Institute for Quality of Healthcare
Radboud University Nijmegen Medical Centre
P.O. Box 9101, 114 IQ healthcare
6500 HB Nijmegen
The Netherlands